Author's response to reviews

Title: Primary pleural leiomyosarcoma with rapid progression and fatal outcomes: a case report

Authors:

ghizlane rais (medghiz@hotmail.com)
soundouss raissouni (soundouss13@hotmail.com)
houda mouzount (hmouzount@hotmail.fr)
meryem aitelhaj (m_meryem_a@hotmail.com)
Siham Khoyaali (s.lkhoyaali@gmail.com)
fadoi El Omrani (elfadwa512@hotmail.fr)
Hind Mrabti (mrabti_h@yahoo.fr)
Ahmed Jelthi (jelthi@menara.ma)
Hassan Errihani (h_errihani@yahoo.fr)

Version: 4 Date: 16 February 2012

Author's response to reviews:

Dear editor

thank you for your interest to our manuscript
the following changes were added according to your suggestions:
- we have corrected the spelling of "Address" in the title page
- we have included the ethnicity "North African" of the patient in both the abstract and case presentation

Thank you